Author's response to reviews

Title: The effect of long-term homocysteine-lowering on carotid intima-media thickness and flow-mediated vasodilation in stroke patients: a randomized controlled trial and meta-analysis

Authors:

Kathleen Potter (pottek04@cyllene.uwa.edu.au)
Graeme J Hankey (gihankey@cyllene.uwa.edu.au)
Daniel J Green (d.j.green@ljmu.ac.uk)
John Eikelboom (eikelbj@mcmaster.ca)
Konrad Jamrozik (konrad.jamrozik@adelaide.edu.au)
Leonard F Arnolda (Leonard.Arnolda@health.wa.gov.au)

Version: 5 Date: 24 August 2008

Author's response to reviews:

Dear Sir/Madam,

Please find attached the final draft of our manuscript, formatted according to the BMC guidelines.

We have altered the sentence in the limitations discussion as suggested by Reviewer 3. The sentence which originally read "Only half of the subjects eligible for inclusion chose to participate" now reads "We collected usable data from only 30% of the subjects originally randomised, approximately half of the subjects identified retrospectively as eligible for inclusion."

Many thanks for accepting this manuscript for publication and don't hesitate to contact me should you require any further assistance.

kind regards,
Kathleen Potter